Author's response to reviews

Title: Chondromodulin-1 directly suppresses growth of human cancer cells

Authors:

Hisashi Mera (his-me@lamen.or.jp)
Hiroyuki Kawashima (inskawa@med.niigata-u.ac.jp)
Tatsuya Yoshizawa (ty2180@columbia.edu)
Osamu Ishibashi (ishibashi@nms.ac.jp)
Moksed Md. Ali (moksedbd@yahoo.com)
Tadashi Hayami (hay@ngt.saiseikai.or.jp)
Hiroshi Kitahara (kita1103@amber.plala.or.jp)
Hiroshi Yamagiwa (Hymgw@med.niigata-u.ac.jp)
Naoki Kondo (uin42727@nifty.com)
Akira Ogose (aogose@med.niigata-u.ac.jp)
Naoto Endo (endless@med.niigata-u.ac.jp)
Hiroyuki Kawashima (kawashim@m2.hinocatv.ne.jp)

Version: 3 Date: 14 May 2009

Author's response to reviews: see over
Dear Sir, J. Ann Le Good, Senior Scientific Editor

Re: “Chondromodulin-1 directly suppresses growth of human cancer cells” by Hisashi Mera and our colleagues; ms. No 2874726222413591.

The attached paper titled “Chondromodulin-1 directly suppresses growth of human cancer cells” has been carefully reviewed by an experienced medical editor whose first language is English and who is specialized in the editing of papers written by physicians and scientists whose native language is not English.

We understood this delay of your response was due to health matter of one of the reviewers. We are grateful to you and the reviewers for the helpful comments on the original version of our manuscript. We have taken all these comments into account and submit again.

We have addressed all the comments by Reviewer 1 and 2, as indicated on the attached pages, and we hope that the explanations and revisions of our work are satisfactory.

We hope that the revised version of our paper is now suitable for publication in the BMC cancer and we look forward to hearing from you at your convenience.

Yours sincerely,

Hisashi Mera, MD

Division of Orthopedic surgery

Niigata University Graduate School of Medical and Dental Science

e-mail: hisme0214@gmail.com

tel: +81-25-227-2272

fax: +81-25-227-0782